Author’s response to reviews

Title: Preclinical Curriculum of Prospective Case-Based Teaching with Faculty- and Student-Blinded Approach

Authors:

Sarah Waliany (swaliany@stanford.edu)
Wendy Caceres (caceres@stanford.edu)
Sylvia Merrell (sylviab@stanford.edu)
Sonoo Thadaney (sonoot@stanford.edu)
Noelle Johnstone (njohnst@stanford.edu)
Lars Osterberg (larso@stanford.edu)

Version: 2 Date: 29 Dec 2018

Author’s response to reviews:

We have added Sylvia Bereknyei Merrell (SBM) and Sonoo Thadaney (ST) in the listed authors' contributions.